Author's response to reviews

Title: Process and effect evaluation of a two-year complex intervention to reduce loneliness in non-institutionalised elderly Dutch people

Authors:

Rianne R Honigh-de Vlaming (rianne.devlaming@wur.nl)
Annemien A Haveman-Nies (annemien.haveman@wur.nl)
Judith H Heinrich (j.heinrich@ggdnog.nl)
Pieter P van 't Veer (pieter.vantveer@wur.nl)
Lisette CPGM de Groot (lisette.degroot@wur.nl)

Version: 2 Date: 3 January 2013

Author's response to reviews:

Dear Editor,

Thank you for giving us the opportunity to revise our manuscript. On your request we included in the methods section a statement about the use of human data. We hope this information meets your requirements.

Yours sincerely,

Rianne Honigh-de Vlaming